Author's response to reviews

Title: Copy Number Variation in Endoglin Locus: Mapping of large deletions in Spanish families with Hereditary Haemorrhagic Telangiectasia type 1

Authors:

Ana Fontalba (afontalba@humv.es)
Jose L Fernandez-Luna (fluna@humv.es)
Roberto Zarrabeitia (roberto.zarrabeitia@scsalud.es)
Lucia Recio (luciarecio@hotmail.com)
Virginia Albiñana (vir_albi_di@yahoo.es)
Maria Luisa Ojeda-Fernandez (mluisa@cib.csic.es)
Carmelo Bernabeu (bernaeu.c@cib.csic.es)
Luis A Alcaraz (uis.alcaraz@bioarray.es)
Luisa M Botella (cibluisa@cib.csic.es)

Version: 10  Date: 4 November 2013

Author's response to reviews:

Dear Editor,

It is for us a pleasure to send the revised versión of our manuscript: MS: 9975711895877967

Copy Number Variation in Endoglin Locus: Mapping of large deletions in Spanish families with Hereditary Haemorrhagic Telangiectasia type 1

Ana Fontalba, Jose L Fernandez-Luna, Roberto Zarrabeitia, Lucia Recio, Virginia Albiñana, Maria Luisa Ojeda-Fernandez, Carmelo Bernabeu, Luis A Alcaraz and Luisa M Botella

We are very grateful to Dr Andreas Lux, by his interest in the manuscript and the detail of his corrections. He is a very good referee. We congratulate him.

Therefore, in the revised version we have included all the comments suggested by Dr Andreas Lux, but number 4. However, we leave this to the best criterium of the editor. If the editor feels that the correction should be done, we will not object.

We hope that the manuscript now is suitable for its publication in BMC Genetics Medicine.

Thank you very much

Best regards,

Luisa María Botella